\documentclass[11pt]{article}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{epstopdf}
\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `dirname #1`/`basename #1 .tif`.png}

\textwidth = 6.5 in
\textheight = 9 in
\oddsidemargin = 0.0 in
\evensidemargin = 0.0 in
\topmargin = 0.0 in
\headheight = 0.0 in
\headsep = 0.0 in
\parskip = 0.2in
\parindent = 0.0in

\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{definition}{Definition}


\begin{document}





\centerline{\bf University of Notre Dame}
\centerline{\bf Department of Mathematics}
\centerline{\bf Spring 1997}
\bigskip
\centerline{\underbar{\bf Information for Students in Math 323}}
\bigskip
\noindent
\underbar{Instructor:}     Professor Kok-Wee Phan, 305 Comp.Math
Building. Phone: 631-7165  E-Mail:  Phan.1$\@$nd.edu
\bigskip
\noindent
\underbar{Text:} \lq\lq Introduction to Probability and its
Applications, 2nd Ed.\rq\rq by Richard L. Scheaffer.
\bigskip
\noindent
You are encouraged to read each section before it is covered.  You are
expected to bring the text to each class.  To do well, you must keep up
with the homework and review frequently.
\bigskip
\noindent
\underbar{Syllabus:}  We expect to cover most of the book.
\bigskip
\noindent
\underbar{Examinations:}  There will be three tests, all at the regular
class time, and a final exam.
$$\align
\text{TEST 1}\quad &\text{Friday, February 7}\\
\text{TEST 2}\quad &\text{Monday, March 3}\\
\text{TEST 3}\quad &\text{Wednesday, April 16}\\
\text{FINAL}\quad &\text{Tuesday, May 6\ (8 - 10 am.)} \endalign$$
Test problems will be multiple choice and will be similar to lecture
examples and homework problems.
\bigskip
\noindent
\underbar{Grading:}  The three tests and final exam will be corrected
and graded uniformly.  Each of the three tests will be worth 100 points,
and the final will be worth 150 points.  Each homework will be graded
and the homework total will then be scaled so that it is worth 50
points.  Letter grades will be based on total points out of the 500
possible.
\bigskip
\noindent
\underbar{Homework:}  Homework will be assigned almost every class
and is due at the end of the next class.  Homework will be accepted one
class period late, but not after that.  Before you begin an assignment
you should carefully work through both the examples done in class and
those in the text.
\bigskip
\noindent
\underbar{Absence from Examinations:}  A student who is absent from
an examination without an official excuse shall receive a grade of zero
for that examination.  A student who is officially excused shall not be
penalized.  If you miss a test for any reason, you should call your
instructor or the Mathematics Department as soon as possible.
\bigskip
\noindent
\underbar{Honor Code:}  Testing will be done under the Honor Code.  On
homework, you are allowed and encouraged to work together, but what
you hand in should not be merely copied.

 \end{document}